
SOAS ISFS Status 15July13
15July13 first look .... edits lost/computer problem, leaving for base and tram trailer extraction.   Last day of project

Re-entering what i recall when looking hour+ ago..... 

     

Personnel jm,cg Kurt, Rudy arrived for teardown

Weather   Forecast is for 40% isolated showers with high in mid 80's.    Rain last night at ~9-11pm CDT (2-
4z) 

Other   ACD operations are over and Tram trailer is going to be pulled today.

System ok Cockpit

  vpn vpn ...   

  df /media/usbdisk %  /scr ..%  Backup .... / ..%  

DSMs ok low,mid,high, gps all ok.  

  ok 
26m 

26m sonic had troubles and continuing since rains last night.   Others appear ok.

WndSpd
/Dir 

ok  

H2O, CO2 ok All Licor7500 diags all remain normal.

TRH ok  

P 43m 43m problems continuing.   

Rad ok  

Tsfc ok 8m tsfc may be having troubles 

Wetness ok  

Soil ok tsoila 4.4cm bad.

VmoteRad ok  

Webcam  

Pond ok 

Sodar    
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